Author’s response to reviews

Title: Reducing the extent of facetectomy may decrease morbidity in failed back surgery syndrome

Authors:

Jingchi Li (lijingchi9405@163.com)
Xiaoyu Zhang (349571476@qq.com)
Wenqiang Xu (xwqshooter@qq.com)
Zhipeng Xi (xizhipeng@vip.126.com)
Lin Xie (xielin6508@163.com)

Version: 3 Date: 17 Jul 2019

Author’s response to reviews:

Dear prof. Darren Byrne and editors of BMC Musculoskeletal Disorders

Firstly, we would like to express our sincerely gratitude to you and other editors of your journal for the potential acceptance of our manuscript after several times of reversion.

The manuscript have been revised according to your demands.

1. The “author contribution” of XYZ have been revised based on your comments (page 15, line 23-24).

2. The etical statement have been revised based on your comments (page 16, line 2)

3. The numbers of ethics have been provided in the page 16 line 3 to meet your requirement.

4. The statement that the volunteer who was provided the imaging data have been repeatedly mentioned in this revised manuscript (Page 4 line 2 and page 16 line 4).
Besides, all revised mentioned above have been changed to red for the clearly identification.

We hope this edition was acceptable.

With kind regards.

Yours

Lin Xie and co-authors of this manuscript.